Author’s response to reviews

Title: Dissecting TSC2-mutated renal and hepatic angiomyolipomas in an individual with ARID1B-associated intellectual disability

Authors:

Bernt Popp (bernt.popp@uk-erlangen.de)
Abbas Agaimy (abbas.agaimy@uk-erlangen.de)
Cornelia Kraus (cornelia.kraus@uk-erlangen.de)
Karl Knaup (KarlXaver.Knaup@uk-erlangen.de)
Arif Ekici (arif.ekici@uk-erlangen.de)
Steffen Uebe (steffen.uebe@uk-erlangen.de)
André Reis (andre.reis@uk-erlangen.de)
Michael Wiesener (michael.wiesener@uk-erlangen.de)
Christiane Zweier (christiane.zweier@uk-erlangen.de)

Version: 2 Date: 25 Jan 2019

Author’s response to reviews:

Dear Editor,

Please find enclosed the second revised version of our manuscript entitled “Dissecting TSC2-mutated renal and hepatic angiomyolipomas in an individual with ARID1B-associated intellectual disability”.

We thank the reviewers for their positive evaluation and helpful comments which we addressed as follows:

Shuli Liang (Reviewer 1): This manuscript was well written, and reported a patient with NDD and renal and hepatic AMLs. Exome and panel sequencing fail to find difference in the ARID1B variant in blood and AMLs, however the TSC2 gene and SIPA1L3 gene mutations were found in
AMLs but not blood. The rare clinical case and post-zygotic variant were adequately analyzed. I think the manuscript can be accepted now.

- We thank the reviewer for the positive evaluation.

We hope that our revised manuscript is now suitable for publication in BMC Cancer.

Yours sincerely,

Christiane Zweier